Author's response to reviews

Title: An intervention to reassure patients about test results in rapid access chest pain clinic: a pilot randomised controlled trial [ISRCTN60618114]

Authors:

Kathryn Hicks (kate.hicks@york.ac.uk)
Kim Cocks (kim.cocks@york.ac.uk)
Belen Corbacho Martin (en.corbacho@york.ac.uk)
Peter Elton (peter.elton@nhs.net)
Anita Macnab (Anita.Macnab@uhsm.nhs.uk)
Wendy Colecliffe (Wendy.Colecliffe@uhsm.nhs.uk)
Gill Furze (gill.furze@coventry.ac.uk)

Version: 2 Date: 14 August 2014

Author's response to reviews:

Details of revisions made to the manuscript in response to the reviewers' comments are in the cover letter_response to reviewer's comments_14.8.14